
\documentclass[twoside,reqno,draft,12pt]{amsart}
\usepackage{amssymb}
\pagestyle{empty}

\setlength{\topmargin}{-.5cm}
\setlength{\headheight}{0cm}
\setlength{\headsep}{0cm}
\setlength{\topskip}{0cm}
\addtolength{\textheight}{4cm}
\setlength{\oddsidemargin}{0cm}
\setlength{\evensidemargin}{0cm}
\addtolength{\textwidth}{4cm}

\newcommand{\ignore}[1]{}

\begin{document}
\thispagestyle{empty}
\noindent Homework 2

\noindent Math 336, Winter `00

\noindent Due Friday, February 4
  
\vskip .2in

\noindent From the textbook:
\begin{itemize}
\item Pages 173-5: 2d (do the problem two ways--once by evaluating the
integral and then taking the limit, and once by taking the limit and 
then evaluating the integral) 5, 6   
\item Pages 181-182: 4, 5  
\end{itemize}

\vskip .2in

\end{document} 




